Author's response to reviews

Title: The assessment of osteoporosis risk factors in Iranian women compared with Indian women

Authors:

Afsaneh Keramat (keramat1@yahoo.com)
Bhushan Patwardhan (bhushan@unipune.ernet.in)
Arvind Chopra (archopra@pn2.vsnl.net.in)
Bagher Larijani (keramat1@yahoo.com)
Ambrish Mithal (ambrishmithal@hotmail.com)
Devlina Chakravarty (keramat1@yahoo.com)
Hossein Adibi (keramat1@yahoo.com)
Ahmad Khosravi (Khosravi2000us@yahoo.com)

Version: 11 Date: 26 February 2008

Author's response to reviews: see over
To: Respected BioMed Central Editorial Production Team

Dear sirs/Madams

Thank you for your email, I have revised the manuscript as follow:

Formatting changes requested
-----------------------------
A) Title - Please use sentence case in the title (capitalise only the first word, and proper nouns), both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

*It is corrected now*

B) Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. We notice Arvind Chopra, Bagher Larijani are in a different order in the submission system to the manuscript.

*It is corrected now*

C) We notice reference 22 is now missing from the manuscript. Please ensure that all the references are cited in the text.

*It is corrected now*

D) We also notice references 41-45 have been cited after the full stop. Please move the reference citations to before the full stop in the sentence.

*It is corrected now*

E) Please change ACKNOWLEDGMENT to Acknowledgments. Please also de-italicise this section.

*It is corrected now*

F) We also notice some unusual references in the text - for example 'table (1)' if you mean the table in reference one, please state 'table in [1]'. If you mean table 1, please state 'Table 1'.

G) Also, to avoid confusion with your results, please cite the references in square brackets.

*It is corrected now*

Many thanks
Afsaneh Keramat